Author's response to reviews

Title: Khat and alcohol use and risky sex behaviour among in-school and out-of-school youth in Ethiopia

Authors:

Derege Kebede (kebeded@whoafr.org)
Atalay Alem (atalayalem@yahoo.com)
Getnet Mitike (getnet_m@yahoo.com)
Fikre Enqusellase (fikreens@yahoo.com)
Frehiwot Berhane (f_berhane@yahoo.com)
Wuleta Lemma (wfrehiwot@yahoo.com)
Yigeremu Abebe (yigeremu.abebe@telecom.net.et)
Reta Ayele (retaayele@yahoo.com)
Tamrat Assefa (tassefa@fhi.org.et)
Tewodros Gebremichael (tmichael@engenderhealth.org)

Version: 3  Date: 30 September 2005

Author's response to reviews:

Dear Mr. David Holmes

Thank you for accepting our manuscript for publication in BMC Public Health. We have made the following editorial changes to the manuscript.

1. Title: We have removed full stop from the end of the title.
2. Abstract: The sub-title Objective has been removed and adjusted according to the format of your journal.
3. The title "Introduction" has been changed to background.
4. References have been cited using numbers in square brackets in the text and the list contains all authors and the term et al has been removed.
5. Tables: The shading and colors within tables have been changed and made bold instead.

I hope it is in an acceptable shape now.

Atalay Alem